Author’s response to reviews

Title: Prescription and dosing of urate-lowering therapy, rather than patient behaviours, are the key modifiable factors associated with targeting serum urate in gout

Authors:

Nicola Dalbeth (n.dalbeth@auckland.ac.nz)
Meaghan E House (m.house@auckland.ac.nz)
Anne Horne (a.horne@auckland.ac.nz)
Keith J Petrie (kj.petrie@auckland.ac.nz)
Fiona M McQueen (f.mcqueen@auckland.ac.nz)
William J Taylor (william.taylor@otago.ac.nz)

Version: 2 Date: 10 September 2012

Author’s response to reviews:

Dear Dr Ding,

Re: Prescription and dosing of urate-lowering therapy, rather than patient behaviours, are the key modifiable factors associated with targeting serum urate in gout.

Thank you for the positive comments from both reviewers. We have changed the title as recommended by Dr Pope (referee 1). It would appear from the reviews provided that no other changes have been requested.

A clean copy of the revised manuscript has been uploaded. A copy with changes tracked is also attached. We hope that this manuscript is now acceptable for publication.

With kind regards,

Yours sincerely,

Nicola Dalbeth